

Dear Editor,

We'd like to thank the reviewers for their careful readings and valuable comments concerning our manuscript entitled "**Hyperglycemic hemianopia: A case report and literature review**" (Manuscript NO: 60321). We believe the constructive feedback will improve the manuscript and increase its potential impact to the community. We read the comments carefully and made revision according to the suggestions. The main corrections in the manuscript and the responds to the reviewer's comments are as followed:

**Responds to the reviewer's comments:**

**Reviewer #1:**

**1. Response to comment:** It would be interesting to know if he was treated in an outpatient basis, or if he was addmitted, for how long did he stay in the hospital.

**Response:** The patient was admitted and stayed in our hospital for two weeks. We have added this information in the "OUTCOME AND FOLLOW-UP" part according to the reviewer's suggestion. "The patient stayed in our hospital for two weeks, and his homonymous hemianopia was gradually relieved."

**2. Response to comment:** It would also be interesting for the authors to further comment on the patient's hyponatremia (considering it has also been described as a cause for hemianopia).

**Response:** Hyponatremia presenting with visual disturbance has also been described in adults. However, in previously reported cases, visual defect was usually accompanied with more typical symptoms and signs such as gastrointestinal upset, significant lethargy, headache, or nausea and vomiting. Our patient, on the contrary, didn't report these symptoms. Besides, hyponatremia may induce cortical blindness through localized occipital lobe edema. While in our patient, cranial MRI showed characteristic T2 and FLAIR subcortical hypointensity. The occipital lesion relatively spared the overlying cortex, which was different from the case in hyponatremia.

**3. Response to comment:** Writing in the english language needs further revision and minor corrections.

**Response:** We are very sorry for the language problem. We have uploaded an edited version of our manuscript.

**4. Response to comment:** I found the language classification was grade C. Please visit the following website for the professional English language edit.

**Response:** We are really sorry that we mistakenly uploaded an unedited manuscript. We have re-upload an edited one.

**5. Response to comment:** I found the authors did not provide the approved grant application form(s). Please upload the approved grant application form(s) or funding agency copy of any approval document(s).

**Response:** We are sorry for the careless omission. We add the approved grant application forms.

**6. Response to comment:** I found the authors did not provide the original figures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

**Response:** We upload a PowerPoint file which includes the original figures.

**7. Response to comment:** The author should number the references in Arabic numerals according to the citation order in the text.

**Response:** We arrange the references in Arabic numerals according to the citation order in the text.

Special thanks to you for your good comments.

Dr. Guohua Zhao

Department of Neurology

The Fourth Affiliated Hospital, Zhejiang University School of Medicine

Yiwu, Zhejiang 322000

China